Title: A randomized placebo-controlled trial of delayed-release dimethyl fumarate in patients with relapsing-remitting multiple sclerosis from East Asia and other countries

Authors:

Takahiko Saida (saida_takahiko@maia.eonet.ne.jp)
Takashi Yamamura (yamamura@ncnp.go.jp)
Takayuki Kondo (takakon78@hotmail.com)
Jang Yun (jang.yun@biogen.com)
Minhua Yang (minhua.yang@biogen.com)
Jie Li (ieli97@yahoo.com)
Lalitha Mahadavan (lalitha.mahadavan@biogen.com)
Bing Zhu (bing.zhu@biogen.com)
Sarah Sheikh (sarah.sheikh@biogen.com)

Version: 4 Date: 01 Nov 2018

Author’s response to reviews:

1 November 2018
Benjamin Ragen
BMC Neurology

Manuscript ID: NURL-D-17-00392R3

Title: A randomized placebo-controlled trial of delayed-release dimethyl fumarate in patients with relapsing-remitting multiple sclerosis from East Asia and other countries
Dear Dr. Ragen,

Thank you for the thorough review. The editor’s suggestions have been addressed as detailed below.

We verify that the references included in the manuscript are accurate.
Please let me know if we missed anything or if you have any additional queries.
Thank you for considering our submission. We look forward to your response.

Yours sincerely,

Takahiko Saida

Takahiko Saida, MD, PhD
Kansai Multiple Sclerosis Centre
Kyoto Min-iren Central Hospital
Nishinokyo-Kasuga-cho 16-44-409
Nakakyo-ku, Kyoto 604-8453, Japan
Phone: 81-(0)75-468-8642
Fax: 81-(0)75-802-2380
E-mail: saida_takahiko@maia.eonet.ne.jp
Technical Comments

1) Please submit a revised manuscript without track changes.
   a) Authors’ response: A clean, revised manuscript has been uploaded.

2) Please include the list of ethics committees as a Supplementary File.
   a) Authors’ response: A list of ethics committees has been added to the supplementary materials, Saida_Supplementary Materials_IRB Tables and this is referenced in the Ethics Approval and Consent to Participate section (pg 26).